Summary of Apr 28 Meeting to Discuss Plans and Schedule Outlook for Commissioning Cryogenics Aspects of the HINS    325 MHz Cavity Test Cave

Bill Soyars reported that the approval to cool down will likely be signed today or tomorrow by Roger Dixon. During the rest of this week, cryo heaters will be wired, purge flow plumbing for the He Utube purge will be installed, and stinging will take place. The cooldown will begin on Monday.

Mike Andrews will check that the ODH level of the cave is unchanged. (Due to the fact that the roof is not in place.)

We currently have one 200W amplifier (from TD/VTS) for CW testing. This power level is not sufficient for SSR2 testing. We will likely order 2 more of the Ophir amplifiers which we currently have. One will be sent back to TD and two will remain at meson. The outputs of the two amplifiers can then be combined to 400 W. These amplifiers cost $10k apiece, so Dave Peterson will spend a week or two searching for off-the-shelf  lower bandwidth(a few MHz), more inexpensive amplifiers. If none exist, we will just buy two more of what we have.
The question was raised as to whether or not the LLRF system would be able to provide for datalogging of raw parameters Pf, Pr, and Pt. Dave verified with Paul after the meeting that this is possible up to 15 Hz. (1 Hz is sufficient for us). An email from Paul said that ‘the data will be provided by the front-end as a raw value and can be scaled by a database transform on the ACNET side which EPICS can probably also do as well.’
